Author’s response to reviews

Title: How parents perceive screen viewing in their 5-6 year old child within the context of their own screen viewing time: a mixed-methods study

Authors:

Janice Thompson (j.thompson.1@bham.ac.uk)
Simon Sebire (simon.sebire@bristol.ac.uk)
Joanna Kesten (jo.kesten@bristol.ac.uk)
Jesmond Zahra (jez.zahra@bristol.ac.uk)
Mark Edwards (mark.edwards@bristol.ac.uk)
Emma Solomon-Moore (emma.solomon@bristol.ac.uk)
Russell Jago (russ.jago@bristol.ac.uk)

Version: 2 Date: 25 Apr 2017

Author’s response to reviews:

We have added the sentence "As part of the informed consent process parents provided written consent for publication for both parent and child data." to the ethics section as requested.